Author's response to reviews

Title: Appendiceal Duplication, A Rare Anatomical Anomaly: a case study

Authors:

Ali Mahmood (StrikerMD@hotmail.com)
Nadia F Mahmood (nfmahmoo@texaschildrens.org)
Jennifer L Williams (JLwilliams@texaschildrens.org)

Version: 2 Date: 20 July 2011

Author's response to reviews:

Dear Editorial Board,

Thank you for responding to our manuscript.

Enclosed are the revisions that were required. They have been entirely implemented.

Thank you again for your rapid initial rapid review and response.

Highest Regards,

Ali Mahmood, M.D.